Author's response to reviews

Title: Reporting of harm in randomized controlled trials evaluating stents for percutaneous coronary intervention

Authors:

MORGANE ME ETHGEN (morganeethgen@hotmail.com)
ISABELLE IB BOUTRON (isabelle.boutron@bch.ap-hop-paris.fr)
PHILIPPE GABRIEL PhGS STEG (gabriel.steg@bch.aphp.fr)
CARINE CR ROY (carine.roy@bch.ap-hop-paris.fr)
PHILIPPE PhR RAVAUD (philippe.ravaud@bch.ap-hop-paris.fr)

Version: 5 Date: 13 January 2009

Author's response to reviews:

As requested and conformly to journal style, we included a section for Authors’ Contributions, a section for Abbreviations and we converted the sections Study design, Data Sources, Study selection and Data extraction in the abstract into a single section for Methods.